Author’s response to reviews

Title: The global switch from trivalent oral polio vaccine (tOPV) to bivalent oral polio vaccine (bOPV): Facts, Experiences and Lessons Learned from the south-south zone; Nigeria, April 2016

Authors:

Enya Bassey Bassey (bassey69@yahoo.com; bbenva69@gmail.com)
Fiona Braka (brakaf@who.int)
Rui Gama Vaz (ruivaz@who.int)
William Komakech (komakech@who.int)
Sisay Gashu Tegegne (gashutegegns@who.int)
Sylvester Toritseju Maleghemi (maleghemis@who.int)
Richard I Koko (kokor@who.int)
Thompson U Igbu (igbut@who.int)
Faith Ireye (ireyef@who.int)
Sylvester O Agwai (agwais@who.int)
Godwin Ubong Akpan (akpang@who.int)
Abdul-Aziz Garba Mohammed (mohammedabd@who.int)
Angela Okocha-Ejeko (peaches_spears@rocketmail.com)

Version: 4 Date: 14 Sep 2017

Author’s response to reviews:

Reviewers Comments: I have one remaining minor comment related to the DRC event that you cited in the first paragraph of the discussion upon my previous suggestion. Recognizing that there is still some scientific debate about the probability that this cVDPV2 relates to pre-switch tOPV use instead of inadvertent post-switch tOPV use, I would suggest you edit the sentence to say "As of June 2017, two separate cases of cVDPV2 possibly associated with inadvertent use of tOPV." Author's Comments: Sentence has been modified to read: “As of June 2017, two separate cases of cVDPV2 possibly associated with inadvertent use of tOPV"